Status Report For TO 025 – Earth Observing 1 Operations
 (03/19/14 – 03/25/14)
Overall Performance
Spacecraft Status
Real-Time

· Maintaining nominal spacecraft state of health 

· Created ATS load to support USN test

· Reviewing options to replace existing OLD CNE computers. 

· Conducted HYP only LCAL along with the Nominal LCAL

· Reviewing the EOMP report

· Reviewing already existing PROCs that can assist with spacecraft power down

· Gathering a list of new PROCs that would need to be created

· The FOT is monitoring the spacecraft SOH post TSM 60 change and all is normal at this point. 

· The FOT plans to change ground limits to coincide with TSM 54, and 60 changes

· Gathering various mnemonics to change with updated limits

· On DOY 070 at 17:49z the FOT performed following activities: 

· Down linked original table 33 (without TSM 60 threshold 1 and 2 changes)

· Uplinked modified table 33 (modified TSM 60 threshold 1 and 2)

· The FOT monitored the TSM 54 change, spacecraft SOH was normal. 

· On DOY 63 at 19:20z the FOT performed the following activities: 

· Down linked original table 33 (without TSM 54 threshold 2 and 4 changes)

· Uplinked modified table 33 (modified TSM 54 threshold 2 and 4)

· Verified the changes and committed it onboard

· Dumped table 33 from RAM to get the TSM 54 and 60 threshold values 

· Reviewed procedures necessary to make TSM threshold changes

· Trending lunar calibrations to see power change overtime

· Designing the added steps to add goal upload automation to 32K TDRS supports

· The current format of products prohibits the automation of 32k tdrs (either all TDRS or none) 

· Determining which format changes would let goal automation be added to automation for TDRS 32k only 

· Testing partial format changes that would allow for TDRS goal automation

· Reviewing x-band power constraints for image and pass scheduling using ASPEN.

· Trending the x-band positive phase current mnemonic
· Mission Planning

· Working with WPS group to help USN certify angle data at South Point, Hawaii and Dongara, Australia

· FDF verifying angle data from the test

· Created the mission plan for HYP Only LCAL and the Nominal Lunar Cal

· Migrated new ASPEN code to ASPEN1 (our primary ASPEN system) 

· Successfully tested the normal operations

· Making changes to MOC scripts for lunar calibrations to reduce manual changes and edits.

· Worked with White Sands to incorporate schedule changes due to wallops x-band data playback testing.

· Working with ASPEN developers to configure various Lunar Calibration sequences 

· Investigating the steps needed to perform an HSI solar cal from an ATS load

· Working with ASPEN developer on solar calibration

· Looking into what would be needed for manual testing of reducing the slew and settling time of the first image of the day

Flight Dynamics 

· Creating an STK scenario to generate view times for a USN test. 

· Created Lunar Calibration products

· Ran EO-1 De-orbit scenarios

· Monitored the changed delivery scripts that send to the new IP address for EDC

· Primary system works fine, backup flight dynamics system needs to be put into the firewall at EDC for the new system there

· Running a 5 year ephemeris to determine MLT times 

Central File Hub

· Phase one tests are ready for outside entity delivery.

· SGS and WGS tracking data are waiting for a break in WALLOPS activities to begin testing.

· Phase 2 FD scripts have no changes from Phase 1 scripts

· All non-FD scripts are implemented Phase 2 

· Beginning Phase 3 work on deliveries for the central file hub

· Creating test file structure for central file hub deliveries

· Tested  password less ssh between FD systems and file hubs successfully

· Tested script to put FD products onto file hubs successfully, modifying files as needed

· Reviewing current deliveries for products, emails, and destinations

· Creating test sample files for deliveries

· Designing delivery code architecture

· Reviewing timing of deliveries

System Administration

· Worked with Ken and Brad Fergus to investigate and resolve problem with Steve’s computer showing up on the FPD KACE blocklist (due to the erroneous reporting of an older version of Adobe Flash on his computer). Also changed Ken’s name to my name as the SA on the list of EO-1 computers in the KACE list.

· Submitted NAMS request for GSFC KACE in order to monitor Code 428 computers via KACE.

· Reviewed Steve Marques’ equipment list containing the EO-1 computers that will be removed from the CNE network by April 8, 2014.  Provided Steve with some updated information for his list.

· Continued preparing for the removal of all CNE Windows XP computers (6 in total).  Worked with Ron to see if the EO-1 Windows XP computers will be able to be upgraded to Windows 7.  Ran the program Windows 7 Upgrade Advisor Report on all 6 computers and generated a report detailing upgrade information for each computer.

· Steve Toal provided specifications and a quote for new laptops for the EO-1 mission.

· Met with the FOT to discuss whether our current computers can successfully upgrade to Windows 7 or not.  Determined some quotes for various laptop options, as well as costs for upgraded software packages such as Filemaker Pro, if needed.

· Began the periodic reboot of the primary string of operational computers (4 computers completed).  A chkdsk/fsck is performed on each system during this routine maintenance.

· Delivered the Jan 2014 monthly backup tape to the Mission Library.

· Performed periodic review of the EO-1 MOC Visitor Log.

· Still waiting for the EDC to implement a new firewall rule for the Waldorf2 computer so that the computer can send data to the EDC if needed.

· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, regional logserver, and MOC CNE computer.

· Created weekly tape.

Data Collection:
Weekly
· Scheduled  142 science Data Collection Events (DCEs) 
· Loss of 6 scenes 
Milestones / Schedules / Deliverables
· Nothing new to report.
Problems
· None
SOARS

· Opened This Week = 0
· Closed = 0
· Total Open = 2
Risks

· Nothing new to report.
Staffing
· Nothing new to report
Certification
· Nothing new to report.
Customer Meetings

· Nothing new to report
Two-Week Event Look
· USN Antenna Angle Testing 

· Continuation of Phase 1/2 FD, waiting on WALLOPS support 

· Moving on to Phase 3 for central file hub, deliveries

